Author's response to reviews

Title: uPA is upregulated by high dose celecoxib in women at increased risk of developing breast cancer

Authors:

Wenyi Qin (qinw@health.missouri.edu)
Weizhu Zhu (zhuw@health.missouri.edu)
John E Hewett (hewetti@health.missouri.edu)
George Rottinghaus (rottinghausg@missouri.edu)
Yin-chieh Chen (yinchieh_chen@yahoo.com)
John T Flynn (john.flynn@jefferson.edu)
Beth Kliethermes (kliethermesb@health.missouri.edu)
Ferdinando Mannello (ferdinando.mannello@uniurb.it)
Edward R Sauter (sautere@health.missouri.edu)

Version: 3 Date: 8 September 2008

Author's response to reviews:

September 8, 2008

RE: uPA is upregulated by high dose celecoxib in women at increased risk of developing breast cancer

Gentlemen:

Enclosed please find a point by point response to each of the Reviewers' comments. The changes are indicated in red text. Please let us know if anything further is required.

Sincerely,

Edward Sauter